Author's response to reviews

Title: The relationship of living arrangements and depressive symptoms among older adults in sub-Saharan Africa

Authors:

Brittany McKinnon (brittany.mckinnon@mail.mcgill.ca)
Sam Harper (sam.harper@mcgill.ca)
Spencer Moore (mooresp@queensu.ca)

Version: 2 Date: 17 November 2012

Author's response to reviews:

17 November 2012

Dear Editor:

Thank you for your response to my recent submission (MS: 1385072848848107) to BMC Public Health. As requested I have made the following changes:
1. The "findings" section in the abstract has been re-titled "results".
2. I have stated how access to the data was obtained in the first paragraph of the methods section

Kind regards,

Brittany